Author’s response to reviews

Title: MAMI: a birth cohort focused on maternal-infant microbiota during early life

Authors:

Izaskun Garcia-Mantrana (igama@iata.csic.es)
Cristina Alcántara (crisalba@iata.csic.es)
Marta Selma-Royo (mselma@iata.csic.es)
Alba Boix-Amorós (albaboix@iata.csic.es)
Majda Dzidic (majda.dzidic@gmail.com)
Jose Gimeno-Alcañiz (jgimeno@iata.csic.es)
Isabel Úbeda-Sansano (isabel.ubeda@gmail.com)
Ignacio Sorribes-Monrabal (isorribesm@gmail.com)
Ramon Escuriet (rescuriet@catsalut.cat)
Fernando Gil Raga (fgil@hospitalmanises.es)
Anna Parra Llorca (annaparrallorca@gmail.com)
Cecilia Martinez-Costa (ceciliam@comv.es)
Maria Carmen Collado (mcolam@iata.csic.es)

Version: 3 Date: 07 Apr 2019

Author’s response to reviews:

Editor Comments:

1. Please relocate the trial registration information presented on the title page to its own subsection at the end of the Abstract, “Trial registration.”

Authors answer: we have relocated the trial registration to the end of the abstract.
2. It is recognized that authors Cristina Alcantara, Marta Selma-Royo, Alba Boix-Amoros, Majda Dzidic, and Jose Gimeno-Alcaniz are not attributed in your Authors contributions subsection of the Declarations. Please take this opportunity to include these authors initials alongside their individual contributions to the study/manuscript.

Authors answer: we have completed the author contributions for the recognized authors.

3. Please note that in accordance with BMC submission guidelines, there should not be a Conclusions subsection in the Abstract of a Study Protocol, but rather a Discussion subsection. To clarify the subsections of the Abstract should be:

   Background
   Methods
   Discussion
   Trial Registration (number and date)

Authors answer: we have removed the headline “conclusions”

4. At this point you may remove the SPIRIT checklist guidelines additional file as this is no longer needed at this point in the Editorial process.

Authors answer: SPIRIT check list has been removed to the final submission

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Authors answer: we have uploaded a clean version